Author's response to reviews

Title: How I do it: transapical cannulation for acute type-A aortic dissection

Authors:

Andrzej W Sosnowski (asosno@gmail.com)
Rajwinder S Jutley (jutley@btinternet.com)
Nicola Masala (nicomasa72@yahoo.it)
Christos Alexiou (christos_alexiou@hotmail.com)
Justiaan Swanevelder (justiaan.swanevelder@uhl.tr.nhs.uk)

Version: 4 Date: 19 January 2008

Author's response to reviews:

1-Different order of authors (i.e. NM before CA)
2-E-mail addresses for each author on the title page.
3-Use of sentence case for title, headings and sub-headings.
4-Elan root valve (Vascutek Ltd, Glasgow, UK) not anymore Kohler Medical Ltd
5-Acknowledgements section with statement of informed consent obtained from the patient
6-changes in references format